Dear Professor Gu,

Re. Letter to the Editor commenting on: Feasibility and safety of “bridging” pancreaticogastrostomy for pancreatic trauma in Landrace pigs

Thank you for the comments on our Letter to the Editor.

We address the comments pointwise:

1. The authors should explain the abbreviations “PD”, “PJ”, and “HPB” in the main text. 
   This has been done.

2. PMID are missing in the reference list.
   This has been corrected.

3. The title page is missing which should include the article title, authorship, institution, ORCID number, supportive foundations, and corresponding author.
   All of the above have been added.

4. The “Abstract” and “Key words” sections are missing.
   These have been added.

5. The “Core tip” section is missing. Please write a summary of no more than 100 words to present the core content of your manuscript, highlighting the most innovative and important findings and/or arguments. Please provide it after the “Key words” section. 
   This has been added.
6. Conflict-of-Interest Disclosure Form and Copyright License Agreement are missing. *These have been added.*

We believe that the article by Dr Feng and colleagues investigates a very important issue in the management of major pancreatic trauma. We believe the quality of the Letter has been improved with the revisions as suggested by the referees.

Yours sincerely,

On behalf of the co-authors

Eduard Jonas  
MB ChB MMed FCS PhD  
Professor and Head: Surgical Gastroenterology  
University of Cape Town and Groote Schuur Hospital